Author's response to reviews

Title: Primary care physician supply and other key determinants of health care utilisation: The case of Switzerland

Authors:

Andre Busato (andre.busato@memcenter.unibe.ch)
Beat Kuenzi (beat.kuenzi@swisspep.ch)

Version: 6 Date: 8 January 2008

Author's response to reviews:

Manuscript changes were made according to Core pre-accept formatting requirements of Jan 8. 08